

# **JGR** Solid Earth

# **RESEARCH ARTICLE**

10.1029/2023JB027769

#### **Key Points:**

- A glacial isostatic adjustment (GIA) model ensemble characterizes the uncertainty of GIA corrections on GRACE-derived water storage trends
- The GIA correction critically impacts the dis-/agreement between GRACE and climate model wetting/drying trends in Eastern Canada
- Spatially coherent regions of agreement between observed trends and climate model predictions have grown substantially over earlier studies

#### Correspondence to:

A. Eicker, annette.eicker@hcu-hamburg.de

#### **Citation:**

Eicker, A., Schawohl, L., Middendorf, K., Bagge, M., Jensen, L., & Dobslaw, H. (2024). Influence of GIA uncertainty on climate model evaluation with GRACE/ GRACE-FO satellite gravimetry data. *Journal of Geophysical Research: Solid Earth*, *129*, e2023JB027769. https://doi. org/10.1029/2023JB027769

Received 8 SEP 2023 Accepted 15 APR 2024

#### **Author Contributions:**

Conceptualization: Annette Eicker, Meike Bagge, Henryk Dobslaw Funding acquisition: Annette Eicker Investigation: Annette Eicker. Lennart Schawohl, Klara Middendorf, Meike Bagge, Laura Jensen Methodology: Annette Eicker, Klara Middendorf, Meike Bagge, Henryk Dobslaw Project administration: Annette Eicker Software: Annette Eicker, Lennart Schawohl, Klara Middendorf. Laura Jensen Supervision: Annette Eicker. Henryk Dobslaw Visualization: Lennart Schawohl, Klara Middendorf Writing - original draft: Annette Eicker, Meike Bagge, Henryk Dobslaw

© 2024. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

# Influence of GIA Uncertainty on Climate Model Evaluation With GRACE/GRACE-FO Satellite Gravimetry Data

Annette Eicker<sup>1</sup>, Lennart Schawohl<sup>1</sup>, Klara Middendorf<sup>1</sup>, Meike Bagge<sup>2,3</sup>, Laura Jensen<sup>2</sup>, and Henryk Dobslaw<sup>2</sup>

<sup>1</sup>HafenCity University Hamburg, Hamburg, Germany, <sup>2</sup>German Research Centre for Geosciences, Potsdam, Germany, <sup>3</sup>Federal Institute for Geosciences and Natural Resources, Hanover, Germany

Abstract Global coupled climate models are in continuous need for evaluation against independent observations to reveal systematic model deficits and uncertainties. Changes in terrestrial water storage (TWS) as measured by satellite gravimetry missions GRACE and GRACE-FO provide valuable information on wetting and drying trends over the continents. Challenges arising from a comparison of observed and modelled water storage trends are related to gravity observations including non-water related variations such as, for example, glacial isostatic adjustment (GIA). Therefore, correcting secular changes in the Earth's gravity field caused by ongoing GIA is important for the monitoring of long-term changes in terrestrial water from GRACE in particular in former ice-covered regions. By utilizing a new ensemble of 56 individual realizations of GIA signals based on perturbations of mantle viscosities and ice history, we find that many of those alternative GIA corrections change the direction of GRACE-derived water storage trends, for example, from gaining mass into drying conditions, in particular in Eastern Canada. The change in the sign of the TWS trends subsequently impacts the conclusions drawn from using GRACE as observational basis for the evaluation of climate models as it influences the dis-/agreement between observed and modelled wetting/drying trends. A modified GIA correction, a combined GRACE/GRACE-FO data record extending over two decades, and a new generation of climate model experiments leads to substantially larger continental areas where wetting/drying trends currently observed by satellite missions coincide with long-term predictions obtained from climate model experiments.

**Plain Language Summary** The satellite gravimetry missions GRACE and GRACE-FO measure changes in the lateral distribution of water stored on Earth. These data give important insight into climate-driven wetting and drying on the continents and can thus be applied to evaluate the output of climate models designed to simulate current and future changes in the Earth system. However, GRACE/-FO are not only sensitive to changes in water storage, but also to other mass change processes such as glacial isostatic adjustment (GIA). GIA refers to the uplift of the land surface, which is still ongoing after the melting of the heavy ice cover from the last ice age. To isolate climate-induced water storage trends, the GIA effect has to subtracted from the GRACE/-FO time series. With 56 different GIA models, we show that the choice of this correction affects the direction of the GRACE-derived water storage trend from wetting to drying (or vice versa). Therefore, it is crucial for the dis-/agreement of the observed trend direction and climate models, particularly in the previously ice-covered regions of Canada. With the revised data processing scheme, the satellite observations document long-term changes in terrestrial water storage in various regions that are in line with climate model projections.

#### 1. Introduction

Readily available freshwater resources are gradually declining in many parts of the world due to increasing demands from human societies for use in agriculture, industry and households (Jasechko & Perrone, 2021). In addition, natural replenishment of those resources is gradually changing under anthropogenic climate change, with shifting precipitation patterns all over the world (Deininger et al., 2020). As a fundamental variable for the interactions between the land-surface and the atmosphere, water kept in the soil or in other hydrological storage compartments has a critical role in the flux of latent, radient, and sensible heat (Mahfouf, 1991). Long-term wetting or drying conditions of terrestrial water storage (*TWS*) in certain regions of the world in consequence of persistent hydrometeorologic changes would carry important information for decision makers concerned with both national and transboundary water management. However, interactively coupled models used to predict future climatic conditions still exhibit a large spread in the representation of soil moisture (Qiao et al., 2022), snow (Mudryk et al., 2020) and other land water related variables (Song et al., 2021). It is therefore important to use a



Writing – review & editing: Annette Eicker, Lennart Schawohl, Klara Middendorf, Laura Jensen diverse set of independent observations for evaluating model performance in order to obtain a realistic view on both capabilities and deficits of numerical models.

The satellite missions Gravity Recovery And Climate Experiment (GRACE (Tapley et al., 2004), in operation from 2002 to 2017) and GRACE-Follow-On (GRACE-FO (Landerer et al., 2020), since 2018) have been observing TWS changes from space with global coverage. Jensen et al. (2019) were the first to demonstrate the potential of this data set for the evaluation of long-term drying and wetting trends in water storage as represented in global coupled climate models taking part in the Coupled Model Intercomparison Project Phase 5 (CMIP5), and Jensen et al. (2020) extended this investigation toward analyzing the temporal evolution of the annual cycle of water stored on ground in model experiments of CMIP6. However, besides the successful identification of "hot spot" regions in which water storage trends in the GRACE data record are likely an indication of a long-term evolution of the global water cycle, Jensen et al. (2019) also revealed several challenges: (a) short observation time spans (only 14 years of data were used for that study) are dominated by inter-annual variations that mask long-term climate-related trends, and (b) discrepancies between observed TWS (full integrated water column, possibly superimposed by non-hydrological mass changes) and modeled terrestrial water storage mTWS (CMIP models provide only soil moisture and snow and therefore do not provide information about all storages that were found to be relevant to explain the GRACE observations) regionally hamper the interpretation of results. Apart from climate model uncertainties, these challenges might be the reason for a disagreement between models and observation on the direction of the trend in several regions, such as in Eastern Canada, the Amazon region, the southern part of Africa, or eastern China (see Jensen et al. (2019), Figure 7). With the length of the data record steadily increasing (20 years of GRACE/-FO observations at the time of writing), limitations imposed by the observation time span will become less critical over the remaining lifetime of GRACE-FO and potential successor missions (Daras et al., 2023; Massotti et al., 2021; Wiese et al., 2022). In addition, a new generation of climate models taking part in the CMIP6 intercomparison can be expected to bring further improvements from the modeling perspective possibly leading to reduced model uncertainties.

However, the task of signal separation persists when vertically integrated gravity observations are to be used for assessing modelled water storage compartments. Jensen et al. (2019, 2020) identified uncertain regions due to anthropogenic impacts on the water cycle, that are typically not taken into account by climate models (e.g., groundwater pumping and surface water management with dams and reservoirs), and surface water storage in natural lakes and wetlands. In addition, GRACE data is also affected by non-hydrological mass transport processes such as earthquakes. The necessary reduction of glacial isostatic adjustment (GIA) effects from the GRACE mass change data was identified as another important source of uncertainty for GRACE-derived *TWS* trends. GIA, primarily manifesting itself in a continued vertical motion of regions that were heavily glaciated during the time of the last glacial maximum approximately 20,000 years ago, is often corrected for in geodetic data analysis. However, models used for this correction carry uncertainties related to the specific treatment of, for example, ice history, mantle rheology, sea level variations, and coastline migration, as it might be deduced from the frequent updates of such correction models that took place in particular during the years right after the launch of the GRACE mission (Tapley et al., 2019). Frequently used GIA correction models include ICE-6D\_D by Argus et al. (2014), the empirical GIA model LM17.3 utilized in Jäggi et al. (2019), or regionally optimized estimations as described by Sasgen et al. (2017).

The issue of GIA uncertainties shall be investigated in this study to answer the following questions: How does the treatment of GIA signals influence (a) the determination of *TWS* trends from GRACE/FO data and (b) the (dis-) agreement between drying/wetting trends in satellite gravimetry and CMIP6 climate models? Is the apparent disagreement between observations and models in the highly GIA-affected region in Canada potentially caused by errors in the GIA reduction?

To investigate these questions, we derive gridded linear trends from the combined GRACE and GRACE-FO *TWS* time series for the time span 2002/04–2021/11 and compare them to long-term trends (1850–2100) of modeled *TWS* (*mTWS*) approximated by the multi-model median (MMMed) trends of the sum of the two variables soil moisture and snow provided by an ensemble of CMIP6 models. Uncertainties of GIA modeling are derived from an ensemble of 56 GIA models that differ both in ice history and Earth structure parametrization (Bagge et al., 2021). To assess the influence of the choice of the GIA correction on the GRACE-derived water storage trends, each of the GIA models is successively subtracted from the GRACE time series to come up with 56 different *TWS* trend estimates. The sensitivity of the sign of the trend to the choice of the GIA model is then





Figure 1. Secular trend in terrestrial water storage (*TWS*) calculated from GRACE/GRACE-FO monthly Level-3 grids for the period 2002/04 until 2021/11 as available from GravIS, where the GIA model by Peltier et al. (2018) has been subtracted.

analyzed for each continental grid cell. Subsequently, for each of these TWS trend maps the agreement of the direction of the trend is compared to the mTWS MMMed CMIP6 trend to assess the relevance of the GIA correction for the comparison of GRACE with climate model output.

## 2. Data

#### 2.1. Terrestrial Water Storage From GRACE/GRACE-FO

We use GRACE and GRACE-FO gridded level 3 *TWS* data from the time span 2002/04 to 2021/11 based on the Release 06 processing of GFZ provided on the GravIS portal (Boergens et al., 2020) mapped to a 2° latitude-longitude global grid. Post-processing steps include the subtraction of the long-term mean field, anisotropic filtering (Horvath et al., 2018; Kusche et al., 2009), the replacement of the coefficients  $c_{10}$ ,  $c_{11}$  and  $s_{11}$  from a model-guided approximation method for the geocenter motion (Swenson et al., 2008), the coefficients  $c_{20}$ ,  $c_{30}$ ,  $c_{21}$  and  $s_{21}$  from estimates based on SLR (Cheng & Ries, 2017; Dahle et al., 2019; Loomis et al., 2020), the correction of the aliased signal of the S2 Tide (Landerer et al., 2020; Stammer et al., 2014), and the correction of signals associated with three major earthquakes (Sumatra-Andaman 2004, Chile 2010, and Tohoku-Oki 2011) (Boergens et al., 2020). To account for the gravity changes induced by GIA, secular rates of gravity change provided by Peltier et al. (2018) were subtracted in order to arrive at gridded estimates of *TWS*. We recall that all those post-processing steps are part of the GRACE-based gridded terrestrial water storage available from GravIS, and will be thus implicitly included in any hydrological application of GRACE observations that is based on data from that website. Therefore, we consider the GIA correction by Peltier et al. (2018) as a reasonable starting point for our study.

For each grid cell, a linear trend is estimated together with the annual and semi-annual harmonics. The trend map is displayed in Figure 1 with purple colors illustrating a long-term drying and green colors indicating a wetting trend. Please note that due to the lack of continental ice dynamics in the CMIP6 models, Greenland and Antarctica are excluded from any further analysis.

#### 2.2. GIA-Induced Gravity Rates

Within the GRACE data processed for the GravIS, GIA is treated as a deterministic correction model that is subtracted without the consideration of its inherent uncertainty, since readily applicable error estimates are typically not available. GIA-induced gravity rates given by Peltier et al. (2018) are based on the ICE-6G\_D ice history and the VM5a radially symmetric viscosity profile in the mantle constrained by a multitude of observational data-sets. The resulting gravity rates are widely used in GRACE/GRACE-FO satellite gravimetry applications (Tapley et al., 2019) and elsewhere in geodesy and geophysics, but in view of sometimes limited observational evidence and various assumptions (e.g., ice history) and simplifications (e.g., spherically symmetric

viscosity distribution in the mantle) applied, those models will naturally represent an imperfect description of the GIA-induced gravity signals.

To further investigate residual uncertainties in GIA-induced gravity rates and to demonstrate the impact of poorly constrained GIA corrections on, for example, climate applications of GRACE/GRACE-FO data, we utilize an ensemble of 56 different GIA realizations from the Viscoelastic Lithosphere and Mantle model (VILMA) (Bagge et al., 2021). Compared to the work by Peltier et al. (2018), the GIA ensemble applied here is advantageous in terms of (a) the utilization of 3D Earth structures considering lateral variations in rheology rather than using radially symmetric viscosity distributions, (b) the consideration of different Earth structure parametrizations as well as (c) the consideration of different ice histories over the last glacial cycle.

The 3D Earth structures used for this work were previously published by Bagge et al. (2020). The viscosity distributions are derived from seismic tomography data from Schaeffer and Lebedev (2013) for depths shallower than 200 km, and from the 2010 update of Grand (2002) for below with a smooth transition and additional geodynamic constraints (e.g., static gravity anomalies, heat flux, mineral physics, Haskell average). The Earth structures account for lateral variations in lithospheric thickness and viscosity in the asthenosphere, upper mantle and transition zone. The upper part of the lower mantle also considers lateral variations, but in the lower part of the lower mantle also set all viscosities lower than  $6 \cdot 10^{19}$  Pa s to this threshold. This allows for larger integration time steps and hence less computing time. The Earth structures differ primarily in the applied conversion algorithm from temperature to viscosity. First, we vary the activation enthalpy factor (*r*) in Arrhenius' law (0.2857, 0.3, 0.4, 0.5, 0.75, 1.0) resulting in a more pronounced lateral variation of the Earth structure and notable differences in lithospheric thickness. Secondly, we alter the radial viscosity profile resulting in different viscosity jumps between upper mantle and transition zone at 410 km depth (s16, sc06, sc06b). The combination of the two types leads to 18 distinctly different Earth structures.

In addition to the various viscosity distributions, we use three distinct ice histories, namely ICE-5G (Peltier, 2004), ICE-6G\_C (Argus et al., 2014; Peltier et al., 2015), and the regionally refined NAICE (Gowan et al., 2016) solution, which is a combination of a regional ice sheet model over North America and ICE-6G\_C for the remaining globe. Since NAICE has significantly less ice over North America than ICE-6G\_C, the residual was uniformly added to the other ice bodies of ICE-6G\_C to ensure that globally integrated ice volume and sea level vary consistently. Although the ICE-5G history has been replaced by newer versions from the same authors, we nevertheless include it into the ensemble to obtain a better understanding about the consequences of possible systematic errors in ice histories, since ICE-5G shows a significant different ice distribution and volume particularly in North America. The 18 different 3D Earth structures are combined with the three different ice histories, resulting in an overall 3D GIA ensemble of 54 unique members, which are complemented by two radially symmetric Earth models, that is, the ICE-6G\_C ice history combined with the 1D Earth structure VM5a, and the ICE-5G ice history combined with the 1D Earth structure VM2, that are widely used elsewhere (Argus et al., 2014; Peltier, 2004; Peltier et al., 2015). In the following analyses, we will thus consider a GIA ensemble with 56 members in total.

The newly constructed ensemble differs from the work of Caron et al. (2018) by utilizing laterally variable 3D Earth structures instead of a vast number of just radially symmetric viscosity profiles. Our approach is closely related to the work of Pan et al. (2022) who similarly utilized an ensemble of GIA models with horizontally and vertically variable Earth structures to assess in particular variations in global mean sea-level since the last glacial maximum. All 56 GIA simulations considered in this study are performed with the VILMA model, that has been applied in many GIA-related studies in the past (Bagge et al., 2021; Dobslaw et al., 2020; Hoening et al., 2023; Huang et al., 2023; Klemann et al., 2008, 2015; Martinec et al., 2018; Schachtschneider et al., 2022). The numerical code solves the sea level equation in the spherical domain and accounts for continuum-mechanics, viscoelasticity, rotational feedback, self-gravitation, mass conservation, and the migration of coastlines. As VILMA accounts for gravitational consistent mass redistributions, we thus directly calculate present-day gravity changes for all ensemble members. Please note that the solution ICE-6G(vm5a) from Peltier et al. (2018) and m\_vm5\_ld\_ice6g from Bagge et al. (2021) use the same 1D Earth structure VM5a as model parametrization, but in detail the ice history differs between ICE-6G\_D (Peltier et al., 2018) and ICE-6G\_C (VILMA). Furthermore, due to model implementation differences in the GIA codes employed, the model results are notably different. Important differences include the impact of compressibility on the Poisson ratio considered by Peltier



(b) Peltier et al. (2018) minus ens. mean



**Figure 2.** Statistical indicators of the VILMA GIA ensemble with 56 members as expressed in EWH: ensemble mean (a), differences of ensemble mean against (Peltier et al., 2018) (b), ensemble spread, that is, difference between the min and max values of the ensemble (c), and ensemble standard deviation (d) for each grid cell (excluding Antarctica).

et al. (2018), the maximum degree of the spherical harmonics expansion within the model integration of 512 versus 170 (VILMA), and the number of iterations for each glacial cycle (three iterations in VILMA), respectively.

Since all the GIA ensemble members considered for this paper are based on a single computational code, systematic errors cannot be addressed in our study. Nevertheless, we note that VILMA has repeatedly participated in model benchmark studies that involved a wide range of different modelling approaches (Martinec et al., 2018; Spada et al., 2011). We therefore consider VILMA as sufficiently mature that robust conclusions can be drawn from the presented experiments. We also note that VILMA incorporates a feedback from changes in the rotational pole on the linear momentum of the viscoelastically deforming solid Earth (Martinec & Hagedoorn, 2014). This VILMA-inherent pole tide (Wahr et al., 2015) has been removed to make the modelling results consistent with the GRACE processing standards. Further details on those recent VILMA improvements will be published in a separate paper.

Gravity rates taken from GIA models can be expressed in terms of equivalent water height (EWH) following Wahr et al. (1998) and the conversion from gravity potential to gridded EWH values was carried out using the software package GROOPS (Mayer-Gürr et al., 2021). We deliberately choose this rather unusual expression of GIA-induced gravity rates in order to tightly connect to the climate model evaluation presented later in this paper. As expected, Figure 2 reveals that the largest GIA signals corresponding to trends as large as several cm/a in EWH are present in North America and Fennoscandia. Additional strong signals in Greenland and Antarctica are not discussed throughout this article due to generally missing continental ice dynamics in CMIP6 climate models.

In regions heavily affected by GIA, the positive trends tend to be stronger for the majority of the ensemble members compared to Peltier et al. (2018), since the differences of Peltier et al. (2018) minus the ensemble mean, are rather negative (Figure 2b). However, it should be noted that this particularly depends on the ice history applied, and some ensemble members also exhibit smaller trends in these areas. The ensemble standard deviation (Figure 2d) has values of up to 0.02 m/a in Canada and the spread (i.e., the difference between the minimum and maximum values per grid cell) reaches values of more than 0.04 m/a (Figure 2c). We emphasize that these values are well within the range of the GIA trend signal itself (Figure 2a).





**Figure 3.** Multi-model median (MMMed) calculated from 17 different CMIP6 climate models for the secular trend in *mTWS* as expressed in equivalent water height (EWH) indicating drying (purple) and wetting (green) conditions (a); and the model consensus characterized by the number of models out of the 17 ensemble members that agree on the sign of the trend for drying (orange) and wetting (blue) conditions, with dark/light colors indicate high/low model consensus, that is, more/less than 13 models agreeing (b).

#### 2.3. Climate Model Data From CMIP6

We use climate model output from the sixth phase of the Coupled Model Intercomparison Project (CMIP6) (Eyring et al., 2016). In CMIP6, the so-called "historical" model experiments cover the time span from 1850 to 2014, and subsequent projections following different "shared socioeconomic pathways (SSP)" are provided from 2015 to 2100. As in Jensen et al. (2020), we choose to concatenate the projection "SSP585" representing the upper end in the range of scenarios (O'Neill et al., 2016) with the historical runs for a continuous time series from 1850 to 2100.

Since there is no *TWS* variable readily provided by the CMIP6 models, an approximation of this quantity is derived from the sum of the variables "soil moisture content (*mrso*)"  $[kg m^{-2}]$  and "surface snow amount (*snw*)"  $[kg m^{-2}]$  in line with our previous work Jensen et al. (2019, 2020). In the following, this sum will be referred to as *modelled TWS* or *mTWS* and the corresponding grids are remapped to a common 2° global resolution. It should be noted that there are systematic differences between *TWS* derived from GRACE and *mTWS*. The *mTWS* includes soil moisture and surface snow but no (deep) groundwater, even though part of the groundwater storage change can be expected to be implicitly contained in the soil moisture variations, and no large (artificial) changes in surface waters. Anthropogenic impacts like massive pumping of groundwater or interference with surface waters as in the construction of dams may result in a significant change in the *TWS* time series determined by GRACE, but has no effect on the *mTWS* time series. However, Jensen et al. (2019, 2020) show that the influence of soil moisture and surface snow on the *TWS* time series is large enough to enable a comparison between *mTWS* and *TWS* despite the systematic differences.

The same ensemble of 17 CMIP6 models as in Jensen et al. (2020) that have provided the relevant variables mrso and snw at the time of writing is also selected for this study, excluding some highly correlated models sharing the same sub-models to avoid biasing the ensemble toward a specific model family. A detailed list of the models that have been excluded can be found in the supplementary material of Jensen et al. (2020) in Figure S2. The linear trend of the *mTWS* time series is estimated for each model individually and subsequently the median of the trend values is calculated for each continental grid cell outside Greenland and Antarctica. It should be noted that some of the 17 models have several different runs (mostly differing by their initial conditions, and rarely also in model parameterization, initialization method, or the applied external forcing) leading to a total of 105 ensemble members. However, when computing the median the individual runs are weighted so that the influence of all models on the median is the same, regardless of how many runs from each model are available. As the computation of the median for each grid cell globally smooths out extremes leading to a reduced range of values in the median compared to individual model trend maps, a rescaling of the median trend map was performed based on its empirical cumulative distribution function as described by Jensen et al. (2020). The resulting map is in the following referred to as the Multi Model Median (MMMed) trend and it is shown in Figure 3a. The purple color indicates the simulation of long-term drying, while the green color stands for the models predicting a wetting trend. Since there are large inter-model differences in the individual trend estimates, following Jensen et al. (2019)



GIA Model Ensemble With 20 Members as Selected Based on a Validation Against Globally Distributed Uplift Rates Obtained From Space Geodetic Data

	Earth structure				RMSE
Ice history	Dimension	AEF	RVP	Model name	[mm/a]
ICE-6G_C	1D	vm5a		m_vm5_1d_ice6g	1.26
NAICE	3D	1.0	s16	m_1.0_s16_3d_naice	1.34
NAICE	3D	1.0	sc06b	m_1.0_sc06b_3d_naice	1.37
NAICE	3D	0.75	s16	m_0.75_s16_3d_naice	1.38
NAICE	3D	1.0	sc06	m_1.0_sc06_3d_naice	1.38
ICE-6G_C	3D	1.0	s16	m_1.0_s16_3d_ice6g	1.40
NAICE	3D	0.75	sc06	m_0.753d_naice	1.40
ICE-6G_C	3D	1.0	sc06b	m_1.0_sc06b_3d_ice6g	1.41
NAICE	3D	0.75	sc06b	m_0.75_sc06b_3d_naice	1.41
NAICE	3D	0.2857	sc06	m_0.2857_sc06_3d_naice	1.42
NAICE	3D	0.5	sc06	m_0.5_sc06_3d_naice	1.42
ICE-6G_C	3D	0.75	s16	m_0.75_s16_3d_ice6g	1.42
NAICE	3D	0.3	sc06	m_0.3_sc06_3d_naice	1.42
NAICE	3D	0.4	sc06	m_0.4_sc06_3d_naice	1.42
NAICE	3D	0.5	s16	m_0.5_s16_3d_naice	1.43
ICE-6G_C	3D	0.75	sc06b	m_0.75_sc06b_3d_ice6g	1.43
NAICE	3D	0.5	sc06b	m_0.5_sc06b_3d_naice	1.45
ICE-6G_C	3D	1.0	sc06	m_1.0_sc06_3d_ice6g	1.45
ICE-6G_C	3D	0.5	s16	m_0.5_s16_3d_ice6g	1.45
NAICE	3D	0.4	s16	m_0.4_s16_3d_naice	1.45

*Note.* A larger activation enthalpy factor (AEF) causes larger lateral viscosity variations and lithospheric thicknesses. The radial viscosity profile (RVP) controls the viscosity jump between the upper mantle and transition zone. AEF and RVP are only applicable for 3D structures. The example models presented in Figure 6 are marked in bold print.

we additionally display the "model consensus" in Figure 3b with dark colors indicating high consensus regions, that is, at least 13 out of 17 models (above 75%) agree on the sign of the drying (orange) or wetting (blue) trend.

#### 3. GIA Influence on TWS Trends From GRACE/GRACE-FO

Before looking specifically into the TWS trends from GRACE, we note that not all GIA realizations considered in this study are equally probable in representing the truth. Present-day GIA-induced gravity rates are the consequence of the viscous response of the solid Earth to the ice history over at least the most recent glacial cycle, and numerous—although sometimes rather imprecise or even contradicting—opportunities exist to validate such models. This includes geologic evidence on relative sea-level in the past, geophysical evidence on the temperature distribution in the mantle, geochemical evidence on mantle composition, and geodetic evidence on present-day crustal deformation. It is beyond the scope of this paper to thoroughly evaluate GIA models and we do not aim to identify a single best-fitting correction model for GIA-induced gravity rates. Instead, we will only attempt to classify the initially 56 GIA realizations in order to arrive at a subset of ensemble members that could be reasonably well considered as being equally probable in representing the truth.

As a first categorization step, we remove all models using the ICE-5G ice history. ICE-5G has been discouraged from further use by its authors for reasons of obvious flaws in the ice distribution that are inconsistent with geological evidence that became only available after its publication. Please note that the very same recommendation also applies to even older ice histories like ICE-4G (Peltier, 2002). In a second step, we utilize globally distributed measurements of present-day uplift rates from various space geodetic techniques contributing to ITRF2020 (Altamimi et al., 2023) to identify 20 ensemble members (Table 1) that fit best to the geodetic record.

trum Potsdam GFZ, Wiley Online Library on [14/05/2024]. See

the Terms and Co

21699356, 2024, 5, Downloaded from https://agupubs.onlinelibrary.wiley.com/doi/10.1029/2023JB027769 by Helmholtz

Table 1



# (a) 56 models ensemble (b) 20 models ensemble

) 10 20 30 40 50 0 5 10 15 20 Number of models in the ensemble causing sign change in TWS trend Number of models in the ensemble causing sign change in TWS trend

**Figure 4.** Change of sign of the GRACE-based *TWS* trend when substituting the GIA correction: number of GIA models when considering the (a) full VILMA ensemble consisting of 56 members, and (b) the 20 best-fitting models only. In the colorbar the White/dark red values in the colorbar refer to no/all GIA models.

Generally speaking, 3D models with higher lateral variability and lithospheric thickness (i.e., higher activation enthalpy factor) for all three viscosity jumps between the upper mantle and transition zone fit better to the observed uplift rates than other Earth structures considered.

We note that the combination of the ICE-6G\_C ice history and the radially symmetric VM5a earth structure reveals the smallest misfit to the geodetic data record. This is not surprising, since this combination has been optimized in view of (an earlier version of) the geodetic data that is now used for evaluation. While somewhat longer time-series are now available, the global distribution of geodetic instruments attached to the crust has not changed dramatically over the past decade, so that systematic errors in areas not considered during the optimization cannot be revealed with this validation approach. Instead, we argue that 3D viscosity structures derived from seismic tomography are more realistic than a simplified radial profile of mantle viscosities. We also mention that the ICE-5G(VM2) combination produces excellent results when validated against geodetic data (rmse = 1.24 mm/a), which would bring it at the top of all models considered despite of the fact being now deprecated based on geological evidence. All the ICE-5G experiments with 3D Earth structures, however, perform considerably worse than their ICE-6G and NAICE counterparts.

We will now assess how exchanging the Peltier et al. (2018) GIA model, which is by default subtracted in the GravIS data set, with alternative GIA corrections from the VILMA ensemble will affect the sign of the *TWS* trend (Figure 4). The darker the color, the higher the number of individual GIA ensemble members that lead to a change in *TWS* trend relative to Peltier et al. (2018) when either all 56 or just the 20 best-fitting alternative GIA corrections are considered. In a relative sense, both subfigures agree surprisingly well with each other. There are areas, as in central Europe, where the sign is not changed by any GIA model at all. In arid areas like Mongolia, the TWS signals observed by GRACE are tiny so that minor changes in the GIA model might already lead to a change in sign, so that those features do not carry any significance.

On the other hand, we also note that a substantial number of GIA ensemble members do indeed change the sign of the GRACE-based *TWS* trend in Canada and the Northern U.S. This area is of particular interest because of the mismatch between GRACE and the CMIP5-based *mTWS* trends as revealed by Jensen et al. (2019). *TWS* from GravIS using the Peltier et al. (2018) model are slightly positive in the Labrador region of Eastern Canada (Figure 1). Larger solid Earth signals from VILMA as given in Figure 2b lead to a sign change in *TWS* with the majority of the GIA ensemble members for this area. If those alternative GIA corrections would be applied, GRACE would, in fact, observe a large-scale drying in Labrador and neighboring regions, instead of wetting as indicated currently by the GravIS data.

The number of models that do change the sign of the *TWS* trend differs greatly across North America. At the coast of Labrador and along the St. Lawrence River, roughly 15 out of the 20 best-fitting VILMA ensemble members would lead to a different sign of the *TWS* trend when replacing Peltier et al. (2018). This number is somewhat smaller in other parts of North America, but still an appreciable number of ensemble members would lead to a different sign of the secular trend. On the one hand, this underlines the importance of the GIA correction for any application of GRACE over the North American continent, which is confirmed by the fact that the influences of GIA and the *TWS* trends are mainly in the same order of magnitude of about  $\pm 4$  cm/a. On the other hand, the





**Figure 5.** Agreement of the GravIS *TWS* trend after subtracting alternative GIA models to the MMMed *mTWS* trend: number of *TWS* trends after subtracting each model of the GIA ensemble that show agreement with the *mTWS* MMMed trend, both for the full ensemble (a) and for the 20 best-fitting GIA models (b). In the colorbar the white and dark red values refer to no/ all GIA models.

results also underline the importance of explicitly accounting for errors in any GIA background model applied, and for properly incorporating the associated GIA uncertainty into the error bars provided by GravIS.

# 4. GIA Influence on GRACE-Based Climate Model Evaluation

In the following, we investigate whether the possible modifications of the GRACE trend by changing the GIA correction will have an influence on the outcome of a climate model validation exercise that is based on the satellite gravity data. Following Jensen et al. (2019), for each grid cell the sign of the GravIS *TWS* trend is compared to the sign of the *mTWS* trend computed as the Multi Model Median (MMMed) of the CMIP6 models. Please note that we are deliberately comparing trends calculated from 250 years of climate model experiments with roughly two decades of satellite observations. As elaborated in detail by Jensen et al. (2019), we hypothesize that climate-driven wetting and drying is already imprinted in the gravity record after a comparably short observation period, and it is our aim to identify hot-spot regions where long-term shifts in the hydrometeorological regime predicted by a majority of coupled climate models are already visible in present-day observations. Such assessments would help to bolster the credibility of climate models in general and their predictions on the future evolution of, for example, precipitation as the main driver for changes in *mTWS*. To assess the influence of the GIA model on this comparison, all GIA models in the ensemble are again subsequently subtracted from the GravIS trend prior to the comparison of the sign.

Figure 5a shows the agreement between the MMMed trend from CMIP6 with 56 different *TWS* trends from GravIS after subtracting each of the VILMA ensemble members instead of Peltier et al. (2018). The darker the color, the more GIA ensemble members from VILMA lead to a sign agreement of the trend between GRACE and the CMIP6 MMMed. It should be noted that both white (0) as well as dark brown (56) indicate no impact of the choice of the GIA model. All hotspot regions identified by Jensen et al. (2019) (as discussed in more detail in the next Section) are fully confirmed independently of the GIA correction considered. Other areas like the northern part of the Amazon catchment or the Southeastern U.S. do not show an agreement with the climate models no matter which GIA model is utilized. In all those areas, the consequences of GIA are too weak to become a relevant contributor to gravity changes monitored by the satellites.

On the other hand, however, we identify quite a number of places in notably GIA-affected regions where only a fraction of the VILMA ensemble members under consideration will align the trends from GRACE with the climate models. This is particularly apparent in Manitoba and Ontario with roughly 30 VILMA ensemble members leading to an agreement, and to a lesser extent also in Labrador, where 10 to 20 GIA ensemble members lead to GRACE-based TWS trends with the same sign as in the MMMed of CMIP6. While globally in 27% of the continental area this agreement is influenced by the choice of GIA correction, this fraction reaches about 65% for the area of Canada.

Figure 5b repeats the same analysis focusing on the 20 best-fitting GIA models, and similarly to Figure 4 the two maps agree quite well with each other. Here 22% of the global land areas and 59% of Canada are affected by the



# Journal of Geophysical Research: Solid Earth

(a) GIA: Peltier et al. 2018



(b) GIA: ensemble member m\_1.0\_s16\_3d\_ice6g



## (c) GIA: ensemble member m\_0.5\_s16\_3d\_ice6g



Figure 6. Left: Agreement of the MMMed mTWS trend compared and the GravIS TWS trend using the different GIA models: (a) Peltier et al. (2018), (b) ensemble member m\_1.0\_s16\_3d\_ice6g, and (c) ensemble member m\_0.5\_s16\_3d\_ice6g. Right: The effect of the substitution of the GIA model on the agreement compared to Peltier et al. (2018).

choice of the GIA model. These numbers are somewhat smaller compared to the full ensemble of 56 models, most likely because of the selection process resulting in a slightly enhanced coherence of the best-fitting models.

# 5. Alternative GIA Corrections in CMIP6 Evaluation

As outlined above, the particular selection of the GIA correction can have a notable influence on GRACE applications in the climate sciences. In the following, we will elaborate this further by selecting two specific realizations out of the 56 VILMA ensemble members. The first realization, called  $m_1.0_s16_3d_ice6g$ , belongs to the "s16" group of Earth structures and has shown the best fit of the present-day uplift rates with observational data in the validation of the 3D models from any of the ICE-6G models considered. The second realization, called  $m_0.5_s16_3d_ice6g$  has the same viscosity contrast between the upper mantle and transition zone ("s16" group) but is characterized by a smaller lithospheric thickness and lower viscosities in the asthenosphere underneath Laurentia. While it has a slightly larger misfit against space geodetic data when evaluated globally, it is still among the 20 best-fitting models. Furthermore, it is one of the models with the largest influence on the GRACE/CMIP6 evaluation and has been chosen here to demonstrate the span of possible impacts of the GIA correction on this comparison.

Both VILMA ensemble members are subsequently applied as GIA corrections to the GRACE data, and the resulting TWS trends are then used as the observational reference to evaluate CMIP6 climate model experiments (Figure 6). A red color signals drying, that is, a negative trend in the climate model MMMed, while a blue color corresponds to wetting, that is, a positive trend. A dark color indicates that the GravIS *TWS* trend confirms the



(a) CMIP5 vs. ITSG-Grace2018 with Peltier et al. (2018), 2002/04-2016/08 m

(b) CMIP6 vs. GravIS with m\_0.5\_s16\_3d\_ice6g, 2002/04-2021/11



**Figure 7.** Agreement on the sign of the long-term trend in *TWS* between numerical climate models and satellite gravimetry: (a) multi-model median from CMIP5 and GRACE TWS grids from ITSG-Grace2018, similar to Jensen et al. (2019), (b) multi-model median from CMIP6 and GRACE/GRACE-FO TWS grids from GravIS. Analysis is limited to grid cells in which at least 75% of models agree on the sign of the trend, cf. Figure 3. For legend see Figure 6.

sign of the MMMed *mTWS* trend, while a light color stands for a disagreement between models and observations. In comparison to using the correction by Peltier et al. (2018), see Figure 6a, we find that with  $m_1.0_s16_3d_ice6g$  most of the GRACE signals south of Hudson Bay now show a drying trend that is also predicted by the climate models (Figure 6b). For a better comparison to the Peltier et al. (2018) correction, the right part of Figure 6b summarizes the impact of substituting the GIA model on the agreement between the CMIP6 and GravIS trends. Here dark green/red areas indicate agreement/disagreement for the alternative GIA model, which was not the case before. With the lower lithospheric thickness and lower viscosities in the asthenosphere in experiment  $m_0.5_s16_3d_ice6g$ , even the *TWS* trends in Labrador become negative and thus align with the CMIP6 models. We also note that changing the GIA correction leads to disagreement between GRACE and climate models at selected places in Russia and also at further isolated locations in Asia and South America. Those pattern are, however, much less spatially coherent than in the Labrador region, and are likely related to generally weak *TWS* trends in those areas so that already very small changes to the GIA correction have an impact on the results from the CMIP6 model evaluation.

Please keep in mind that coincidence with CMIP6 shall not be used as a selection criterion for GIA corrections, which still need to be selected based on validation with external data from both modern instrumental records as well as geological evidence. Furthermore, in this analysis we did not mask out regions of low model consensus (cf. Figure 3), therefore the comparison might be also somewhat influenced by uncertainties in the MMMed trend estimate. Nevertheless, the analysis underlines the sensitivity of GRACE applications to GIA particular in North America and other previously and currently glaciated regions, and consequently the importance of an accurate GIA correction for reliable estimates of water storage and ice mass changes.

#### 6. Improvements Since CMIP5 Assessment by Jensen et al. (2019)

As a final note, we show a comparison of the GRACE/GRACE-FO-based CMIP6 model assessment using alternative GIA corrections with the results previously obtained by Jensen et al. (2019) with CMIP5 and a much shorter gravity record (Figure 7a). Those results were based on a *TWS* trend estimation from only 14 years GRACE data from ITSG-Grace2018 (Kvas et al., 2019; Mayer-Gürr et al., 2018) and 21 sufficiently uncorrelated climate model experiments from the CMIP5 archive. For the analysis of hot spot regions, the comparison is limited to grid cells with a high model consensus, that is, a large number of models agreeing on the sign of the trend. In a slight deviation from Jensen et al. (2019), we define 75% instead of 71% of the ensemble members as "high consensus" to be consistent with the smaller CMIP6 ensemble used in the present study (cf. Figure 3).

Figure 7b shows the agreement in sign of the long-term trend between CMIP6 and the longer time series of *TWS* products from GravIS using the GIA model m\_0.5\_s16\_3d\_ice6g. We note a substantial increase of coherent regions where models and observations agree on the trend when compared to the previous assessment with the agreement area having increased from 48.0% for the CMIP5 comparison to 59.1% for the CMIP6 comparison. Most notable, the Mediterranean region and large swaths of central and eastern Europe show a drying climate in the GRACE records that is also predicted by the majority of CMIP6 models. We also see a coherent



21699356, 2024, 5, Downloaded

schu

com/doi/10.1029/2023JB027769 by Heli

on [14/05/2024]. See

Wiley Onlin

OA articles are

applicable Creative

pattern of wetter climate conditions in central Africa that has not been detected in our previous work. This increase in coherence among models and satellite data documents the progress that has been recently achieved both in terms of reducing systematic errors and noise in the satellite gravity data as well as in refining numerical climate models. The latter is underlined by a better consensus of the model ensemble regarding the direction of the trend with 43.7% of the continental area being identified as "high consensus" regions for the CMIP6 ensemble versus only 34.5% for CMIP5.

#### 7. Conclusions

Correcting secular changes in the Earth's gravity field caused by ongoing Glacial Isostatic Adjustment (GIA) are not only important for the quantification of continental ice-mass loss in Greenland and Antarctica, but also for the monitoring of long-term changes in terrestrial water storage from GRACE/GRACE-FO in particular in the former ice-covered regions of North America and Fennoscandia. By utilizing a new ensemble of 56 individual realizations of the GIA signals with the viscoelastic Earth model VILMA using 18 different 3D Earth structures, three different ice histories and two standard 1D models, we find that many of those alternative GIA corrections are able to change the direction of GRACE-derived *TWS* trends, for example, from showing mass gain into drying conditions in particular in Western Canada. The sensitivity of GRACE to GIA uncertainties underlines the importance of continued GIA research required to further improve the understanding of this important geodynamic process, and subsequently to increase the reliability of GIA corrections applied to GRACE and other geodetic data. The change in the sign of the *mTWS* trends induced by an alternative GIA correction subsequently also impacts the conclusions to be drawn when using GRACE as observational basis for the evaluation of climate models as it influences the dis-/agreement between observed and modelled wetting/drying trends. We recall, however, that any decision about the choice of a suitable GIA correction should be based on observational constraints and geophysical expert knowledge and must not be rooted on a better fit with numerical climate models.

We select two particular alternative GIA realizations from VILMA in combination with a newly processed mass trend map from GRACE and GRACE-FO to evaluate these wetting and drying trends in climate model experiments collected by the latest international climate model intercomparison project (CMIP6). Compared to our previous assessment based on CMIP5 and a somewhat shorter GRACE record, we find that the spatially coherent regions of agreement between observations and climate models have grown substantially. This is both related to improvements in the climate model experiments, where in particular deeper soil schemes have led to a considerable more realistic mTWS representation, a combined GRACE/GRACE-FO gravity time-series of roughly 20 years that better allows discriminating long-term trends from interannual-to-decadal climate variability, and also to refined GIA corrections, that lead to a better agreement between GRACE/GRACE-FO and the climate models particularly in the Labrador region. Achievements from very different branches of the geosciences (i.e., refined processing of GRACE/GRACE-FO satellite sensor data, revised upper mantle viscosities, more elaborate soil moisture schemes) eventually lead to a better correspondence of observed and modelled trends in *mTWS*, thereby providing (a) further credibility to the predictive skill of numerical climate models, and (b) further empirical evidence that ongoing changes in the Earth's climate due to continuously increasing concentrations of carbon dioxide and other Greenhouse gases do have a profound (and measurable) impact on hydrological storages already today.

#### **Data Availability Statement**

The GRACE/GRACE-FO data used in the present study are publicly available at the GravIS portal http://gravis. gfz-potsdam.de. The ensemble of GIA models from VILMA is published in Bagge et al. (2023). We acknowledge the World Climate Research Programme's Working Group on Coupled Modelling, which is responsible for CMIP, and we thank the climate modeling groups for producing and making available their model output, for example, at https://esgf-data.dkrz.de/search/cmip6-dkrz/.

#### References

Altamimi, Z., Rebischung, P., Collilieux, X., Métivier, L., & Chanard, K. (2023). Itrf2020: An augmented reference frame refining the modeling of nonlinear station motions. *Journal of Geodesy*, 97(5), 47. https://doi.org/10.1007/s00190-023-01738-w

Argus, D. F., Peltier, W. R., Drummond, R., & Moore, A. W. (2014). The Antarctica component of postglacial rebound model ICE-6G\_C (VM5a) based on GPS positioning, exposure age dating of ice thicknesses, and relative sea level histories. *Geophysical Journal International*, 198(1), 537–563. https://doi.org/10.1093/gji/ggu140

#### Acknowledgments

This study was funded by Deutsche Forschungsgemeinschaft (DFG) within the Research Unit NEROGRAV under the grant EI 772/3-2. MB received funding from the German climate modeling project PalMod (FKZ: 01LP1918A) supported by the German Federal Ministry of Education and Research (BMBF) as a Research for Sustainability initiative (FONA). Additional support has been provided by the European Space Agency via the Science Phase of GRACE-FO within ESA's Third Party Mission Program (Contract no. 4000135530). The authors would like to thank the editors for handling the manuscript and Dr. Holger Steffen and an anonymous reviewer for their constructive feedback. Open Access funding enabled and organized by Projekt DEAL.



- Bagge, M., Klemann, V., Dill, R., & Dobslaw, H. (2023). Present-day gravity changes induced by glacial isostatic adjustment from ensemble simulations with 3d Earth structures. GFZ Data Services. https://doi.org/10.5880/GFZ.1.3.2023.002
- Bagge, M., Klemann, V., Steinberger, B., Latinović, M., & Thomas, M. (2020). 3D Earth structures for glacial-isostatic adjustment models. GFZ Data Services. https://doi.org/10.5880/GFZ.1.3.2020.004
- Bagge, M., Klemann, V., Steinberger, B., Latinović, M., & Thomas, M. (2021). Glacial-isostatic adjustment models using geodynamically constrained 3D Earth structures. *Geochemistry, Geophysics, Geosystems*, 22(11). https://doi.org/10.1029/2021GC009853
- Boergens, E., Dobslaw, H., Dill, R., Thomas, M., Dahle, C., Murböck, M., & Flechtner, F. (2020). Modelling spatial covariances for terrestrial water storage variations verified with synthetic GRACE-FO data. GEM - International Journal on Geomathematics, 11(1), 1–25. https://doi. org/10.1007/s13137-020-00160-0
- Caron, L., Ivins, E. R., Larour, E., Adhikari, S., Nilsson, J., & Blewitt, G. (2018). Gia model statistics for grace hydrology, cryosphere, and ocean science. *Geophysical Research Letters*, 45(5), 2203–2212. https://doi.org/10.1002/2017GL076644
- Cheng, M., & Ries, J. (2017). The unexpected signal in GRACE estimates of C<sub>20</sub>. Journal of Geodesy, 91(8), 897–914. https://doi.org/10.1007/s00190-016-0995-5
- Dahle, C., Murböck, M., Flechtner, F., Dobslaw, H., Michalak, G., Neumayer, K., et al. (2019). The GFZ GRACE RL06 monthly gravity field time series: Processing details and quality assessment. *Remote Sensing*, 11(18), 2116. https://doi.org/10.3390/rs11182116
- Daras, I., March, G., Pail, R., Hughes, C., Braitenberg, C., Güntner, A., et al. (2023). Mass-change and geosciences international constellation (magic) expected impact on science and applications. *Geophysical Journal International*, 236(3), ggad472. https://doi.org/10.1093/gji/ ggad472
- Deininger, M., McDermott, F., Cruz, F. W., Bernal, J. P., Mudelsee, M., Vonhof, H., et al. (2020). Inter-hemispheric synchroneity of holocene precipitation anomalies controlled by Earth's latitudinal insolation gradients. *Nature Communications*, 11(1), 5447. https://doi.org/10.1038/ s41467-020-19021-3
- Dobslaw, H., Dill, R., Bagge, M., Klemann, V., Boergens, E., Thomas, M., et al. (2020). Gravitationally consistent mean barystatic sea level rise from leakage-corrected monthly grace data. *Journal of Geophysical Research: Solid Earth*, 125(11), e2020JB020923. https://doi.org/10.1029/ 2020jb020923
- Eyring, V., Bony, S., Meehl, G. A., Senior, C. A., Stevens, B., Stouffer, R. J., & Taylor, K. E. (2016). Overview of the coupled model intercomparison project phase 6 (CMIP6) experimental design and organization. *Geoscientific Model Development*, 9(5), 1937–1958. https://doi. org/10.5194/gmd-9-1937-2016
- Gowan, E. J., Tregoning, P., Purcell, A., Montillet, J.-P., & McClusky, S. (2016). A model of the western laurentide ice sheet, using observations of glacial isostatic adjustment. *Quaternary Science Reviews*, 139, 1–16. https://doi.org/10.1016/j.quascirev.2016.03.003
- Grand, S. P. (2002). Mantle shear–wave tomography and the fate of subducted slabs. *Philosophical Transactions of the Royal Society of London, Series A: Mathematical, Physical and Engineering Sciences, 360*(1800), 2475–2491. https://doi.org/10.1098/rsta.2002.1077
- Hoening, D., Willeit, M., Calov, R., Klemann, V., Bagge, M., & Ganopolski, A. (2023). Multistability and transient response of the Greenland ice sheet to anthropogenic CO<sub>2</sub> emissions. *Geophysical Research Letters*, 50(6), e2022GL101827. https://doi.org/10.1029/2022GL101827
- Horvath, A., Murböck, M., Pail, R., & Horwath, M. (2018). Decorrelation of GRACE time variable gravity field solutions using full covariance information. *Geosciences*, 8(9), 323. https://doi.org/10.3390/geosciences8090323
- Huang, P., Steffen, R., Steffen, H., Klemann, V., Wu, P., van der Wal, W., et al. (2023). A commercial finite element approach to modelling Glacial Isostatic Adjustment on spherical self-gravitating compressible Earth models. *Geophysical Journal International*, 235(3), 2231–2256. https://doi.org/10.1093/gji/ggad354
- Jäggi, A., Weigelt, M., Flechtner, F., Güntner, A., Mayer-Gürr, T., Martinis, S., et al. (2019). European gravity service for improved emergency management (EGSIEM)—From concept to implementation. *Geophysical Journal International*, 218(3), 1572–1590. https://doi.org/10.1093/ gji/ggz238
- Jasechko, S., & Perrone, D. (2021). Global groundwater wells at risk of running dry. Science, 372(6540), 418–421. https://doi.org/10.1126/ science.abc2755
- Jensen, L., Eicker, A., Dobslaw, H., & Pail, R. (2020). Emerging changes in terrestrial water storage variability as a target for future satellite gravity missions. *Remote Sensing*, 12(23), 3898. https://doi.org/10.3390/rs12233898
- Jensen, L., Eicker, A., Dobslaw, H., Stacke, T., & Humphrey, V. (2019). Long-term wetting and drying trends in land water storage derived from GRACE and CMIP5 models. Journal of Geophysical Research: Atmospheres, 124(17–18), 9808–9823. https://doi.org/10.1029/2018JD029989
- Klemann, V., Heim, B., Bauch, H. A., & Sebastian Wetterich, T. O. (2015). Sea-level evolution of the Laptev Sea and the East Siberian Sea since the last glacial maximum. Arktos.
- Klemann, V., Martinec, Z., & Ivins, E. R. (2008). Glacial isostasy and plate motion. Journal of Geodynamics, 46(3), 95–103. https://doi.org/10. 1016/j.jog.2008.04.005
- Kusche, J., Schmidt, R., Petrovic, S., & Rietbroek, R. (2009). Decorrelated GRACE time-variable gravity solutions by GFZ, and their validation using a hydrological model. *Journal of Geodesy*, 83(10), 903–913. https://doi.org/10.1007/s00190-009-0308-3
- Kvas, A., Behzadpour, S., Ellmer, M., Klinger, B., Strasser, S., Zehentner, N., & Mayer-Gürr, T. (2019). Itsg-grace2018: Overview and evaluation of a new grace-only gravity field time series. *Journal of Geophysical Research: Solid Earth*, 124(8), 9332–9344. https://doi.org/10.1029/ 2019jb017415
- Landerer, F. W., Flechtner, F. M., Save, H., Webb, F. H., Bandikova, T., Bertiger, W. I., et al. (2020). Extending the global mass change data record: GRACE follow-on instrument and science data performance. *Geophysical Research Letters*, 47(12). https://doi.org/10.1029/ 2020GL088306
- Loomis, B. D., Rachlin, K. E., Wiese, D. N., Landerer, F. W., & Luthcke, S. B. (2020). Replacing GRACE/GRACE-FO with satellite laser ranging: Impacts on Antarctic ice sheet mass change. *Geophysical Research Letters*, 47(3). https://doi.org/10.1029/2019GL085488
- Mahfouf, J.-F. (1991). Analysis of soil moisture from near-surface parameters: A feasibility study. Journal of Applied Meteorology and Climatology, 30(11), 1534–1547. https://doi.org/10.1175/1520-0450(1991)030<1534:aosmfn>2.0.co;2
- Martinec, Z., & Hagedoorn, J. (2014). The rotational feedback on linear-momentum balance in glacial isostatic adjustment. *Geophysical Journal International*, 199(3), 1823–1846. https://doi.org/10.1093/gji/ggu369
- Martinec, Z., Klemann, V., van der Wal, W., Riva, R. E. M., Spada, G., Sun, Y., et al. (2018). A benchmark study of numerical implementations of the sea level equation in GIA modelling. *Geophysical Journal International*, 215(1), 389–414. https://doi.org/10.1093/gji/ggy280
- Massotti, L., Siemes, C., March, G., Haagmans, R., & Silvestrin, P. (2021). Next generation gravity mission elements of the mass change and geoscience international constellation: From orbit selection to instrument and mission design. *Remote Sensing*, 13(19), 3935. https://doi.org/10. 3390/rs13193935
- Mayer-Gürr, T., Behzadpour, S., Eicker, A., Ellmer, M., Koch, B., Krauss, S., et al. (2021). Groops: A software toolkit for gravity field recovery and GNSS processing. *Computers & Geosciences*, 155, 104864. https://doi.org/10.1016/j.cageo.2021.104864



- Mayer-Gürr, T., Behzadpur, S., Ellmer, M., Kvas, A., Klinger, B., Strasser, S., & Zehentner, N. (2018). ITSG-Grace2018—Monthly and daily gravity field solutions from GRACE [Dataset]. https://doi.org/10.5880/icgem.2018.003
- Mudryk, L., Santolaria-Otín, M., Krinner, G., Ménégoz, M., Derksen, C., Brutel-Vuilmet, C., et al. (2020). Historical northern hemisphere snow cover trends and projected changes in the CMIP6 multi-model ensemble. *The Cryosphere*, 14(7), 2495–2514. https://doi.org/10.5194/tc-14-2495-2020
- O'Neill, B. C., Tebaldi, C., van Vuuren, D. P., Eyring, V., Friedlingstein, P., Hurtt, G., et al. (2016). The scenario model intercomparison project (ScenarioMIP) for CMIP6. *Geoscientific Model Development*, 9(9), 3461–3482. https://doi.org/10.5194/gmd-9-3461-2016
- Pan, L., Milne, G. A., Latychev, K., Goldberg, S. L., Austermann, J., Hoggard, M. J., & Mitrovica, J. X. (2022). The influence of lateral Earth structure on inferences of global ice volume during the last glacial maximum. *Quaternary Science Reviews*, 290, 107644. https://doi.org/10. 1016/j.quascirev.2022.107644
- Peltier, W. R. (2002). Global glacial isostatic adjustment: Palaeogeodetic and space-geodetic tests of the ICE-4G (VM2) model. Journal of Quaternary Science: Published for the Quaternary Research Association, 17(5–6), 491–510. https://doi.org/10.1002/jqs.713
- Peltier, W. R. (2004). Global glacial isostasy and the surface of the ice-age Earth: The ICE-5G (VM2) model and GRACE. Annual Review of Earth and Planetary Sciences, 32(1), 111–149. https://doi.org/10.1146/annurev.earth.32.082503.144359
- Peltier, W. R., Argus, D. F., & Drummond, R. (2015). Space geodesy constrains ice age terminal deglaciation: The global ice-6g\_c (VM5A) model. Journal of Geophysical Research: Solid Earth, 120(1), 450–487. https://doi.org/10.1002/2014jb011176
- Peltier, W. R., Argus, D. F., & Drummond, R. (2018). Comment on "An Assessment of the ICE-6G\_C (VM5a) Glacial Isostatic Adjustment Model" by Purcell et al. Journal of Geophysical Research: Solid Earth, 123(2), 2019–2028. https://doi.org/10.1002/2016JB013844
- Qiao, L., Zuo, Z., & Xiao, D. (2022). Evaluation of soil moisture in CMIP6 simulations. Journal of Climate, 35(2), 779-800. https://doi.org/10. 1175/jcli-d-20-0827.1
- Sasgen, I., Martín-Español, A., Horvath, A., Klemann, V., Petrie, E. J., Wouters, B., et al. (2017). Joint inversion estimate of regional glacial isostatic adjustment in Antarctica considering a lateral varying Earth structure (ESA STSE Project REGINA). Geophysical Journal International, 211(3), 1534–1553. https://doi.org/10.1093/gji/ggx368
- Schachtschneider, R., Saynisch-Wagner, J., Klemann, V., Bagge, M., & Thomas, M. (2022). An approach for constraining mantle viscosities through assimilation of palaeo sea level data into a glacial isostatic adjustment model. *Nonlinear Processes in Geophysics*, 29(1), 53–75. https:// doi.org/10.5194/npg-29-53-2022
- Schaeffer, A. J., & Lebedev, S. (2013). Global shear speed structure of the upper mantle and transition zone. *Geophysical Journal International*, 194(1), 417–449. https://doi.org/10.1093/gji/ggt095
- Song, X., Wang, D.-Y., Li, F., & Zeng, X.-D. (2021). Evaluating the performance of cmip6 Earth system models in simulating global vegetation structure and distribution. *Advances in Climate Change Research*, *12*(4), 584–595. https://doi.org/10.1016/j.accre.2021.06.008
- Spada, G., Barletta, V. R., Klemann, V., Riva, R. E. M., Martinec, Z., Gasperini, P., et al. (2011). A benchmark study for glacial isostatic adjustment codes. *Geophysical Journal International*, 185(1), 106–132. https://doi.org/10.1111/j.1365-246X.2011.04952.x
- Stammer, D., Ray, R. D., Andersen, O. B., Arbic, B. K., Bosch, W., Carrère, L., et al. (2014). Accuracy assessment of global barotropic ocean tide models. *Reviews of Geophysics*, 52(3), 243–282. https://doi.org/10.1002/2014RG000450
- Swenson, S., Chambers, D., & Wahr, J. (2008). Estimating geocenter variations from a combination of GRACE and ocean model output. Journal of Geophysical Research, 113(B8). https://doi.org/10.1029/2007JB005338
- Tapley, B. D., Bettadpur, S., Ries, J. C., Thompson, P. F., & Watkins, M. M. (2004). GRACE measurements of mass variability in the Earth system. Science (New York, N.Y.), 305(5683), 503–505. https://doi.org/10.1126/science.1099192
- Tapley, B. D., Watkins, M. M., Flechtner, F., Reigber, C., Bettadpur, S., Rodell, M., et al. (2019). Contributions of grace to understanding climate change. *Nature Climate Change*, 9(5), 358–369. https://doi.org/10.1038/s41558-019-0456-2
- Wahr, J., Molenaar, M., & Bryan, F. (1998). Time variability of the Earth's gravity field: Hydrological and oceanic effects and their possible detection using grace. *Journal of Geophysical Research*, 103(B12), 30205–30229. https://doi.org/10.1029/98jb02844
- Wahr, J., Nerem, R. S., & Bettadpur, S. V. (2015). The pole tide and its effect on grace time-variable gravity measurements: Implications for estimates of surface mass variations. Journal of Geophysical Research: Solid Earth, 120(6), 4597–4615. https://doi.org/10.1002/2015jb011986
- Wiese, D. N., Bienstock, B., Blackwood, C., Chrone, J., Loomis, B. D., Sauber, J., et al. (2022). The mass change designated observable study: Overview and results. *Earth and Space Science*, 9(8), e2022EA002311. https://doi.org/10.1029/2022ea002311